Author's response to reviews

Title: Influenza vaccination coverage rates among adults before and after the 2009 influenza pandemic and the reasons for non-vaccination in Beijing, China: A cross-sectional study

Authors:

Shuangsheng Wu (wushuangsheng0606@126.com)
Peng Yang (clairecaoyang@163.com)
Haiyue Li (0418012058@163.com)
Chunna Ma (ma_chun_na@126.com)
Yi Zhang (zps347@163.com)
Quanyi Wang (bjcdcxm@126.com)

Version: 12 Date: 25 June 2013

Author's response to reviews: see over
Dear Editors:
My manuscript, ‘Influenza vaccination coverage rates among adults before and after the 2009 influenza pandemic and the reasons for non-vaccination in Beijing, China: a cross-sectional study’, was revised according to your comments, and was edited by using a professional language editing service.
Many thanks for your suggestion. Thanks very much again for your attention to our paper.
Sincerely yours,
Shuangsheng Wu

For your guidance, a point-to-point response to your comments is appended below.
Your manuscript has now been peer reviewed. However, before we can proceed, we will need you to make some changes to your manuscript. We strongly urge you to make these changes promptly, as we cannot proceed to the next process until we have received a version containing the changes.

1. Further consideration of your manuscript is conditional on improvement of the English used – please bear in mind that as we are a free-access publisher, we cannot bear the costs of copyediting English ourselves. Please ensure particular attention is paid to the abstract. You should have a native English speaking colleague help you with this, if possible, or you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/info/authors/authorfaqs#12.

We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/info/authors/authorfaqs#12.

We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/info/authors/authorfaqs#12.

My manuscript has been revised according to your comments, and was edited by using a professional language editing service. Some language corrections were made, and were marked red in my revised manuscript.

2. We would recommend that you shorten the text in order to improve the clarity of your manuscript and reduce any repetitiveness.
We have shortened the text in the previous versions according to the reviewers’ comments. Additionally, in current version, the typographical/grammatical/editorial errors were corrected to improve the clarity of the manuscript, and some sentences were edited and condensed to reduce the repetitiveness. All these changes were marked red in my revised manuscript.